Author’s response to reviews

Title: Serious postoperative complications induced by medical glue: three case reports

Authors:

Wei Li (weili8308@gmail.com; weili8308@126.com; weili8308@jlu.edu.cn)
Mingming Xiao (xiao623245@163.com)
Yujia Chen (yujiachen129@163.com)
Jiaxing Yang (jiaxingyang@jlu.edu.cn)
Donghui Sun (sundonghui131@163.com)
Jian Suo (suojian0066@126.com)
Daguang Wang (daguangjida@163.com)

Version: 5 Date: 09 Dec 2019

Author’s response to reviews:

Dear Editor,

We are truly grateful to the positive and constructive comments as well as thoughtful suggestions shared with regard to our manuscript entitled "Serious postoperative complications induced by fibrin glue: three case reports" (Manuscript# BMGE-D-18-00579R4). We undertook a careful revision of our manuscript, an incorporated all of the suggested changes. We have submitted our point-by-point responses to the reviewers’ comments and questions below this letter.

We would like to express our deep appreciation for your review of our paper, and hope the revised manuscript is found suitable for publication in your esteemed journal.
Sincerely yours,

Daguang Wang MD, PhD
Department of Gastrointestinal Surgery
The First Hospital of Jilin University
Changchung, Jilin 130021
E-mail: daguangjida@163.com

Editor Comments:

1. The first sentence in the abstract: Various types of medical glues/adhesives/topical coagulants' which we will refer to as medical glues (MG) and have widespread application as surgical adhesives.... should be changed to something like 'Various types of medical glues/adhesives/topical coagulants'(refered to as MG hereinafter) have widespread application as surgical adhesives,'. **

Response: Thank you for this suggestion. We have revised the manuscript to include this change.